Author's response to reviews

Title: Management Of Arrhythmogenic Right Ventricular Dysplasia In Pregnancy: A Rare Case

Authors:

Nilgün Güdücü (nilgun.kutay@gmail.com)
Salih S Kutay (serdarkutay@hotmail.com)
Ebru Özenç (eozenc@hotmail.com)
Çavlan Çiftçi (cavlanciftci@superonline.com)
Alin B Yiiter (alinbasgul@gmail.com)
Herman İşçi (hermani@superonline.com)

Version: 3 Date: 7 January 2011

Author's response to reviews:

Dear Editor,

We have made the changes as suggested by the reviewers:

-Third paragraph: In the third trimester she had palpitations and became symptomatic again.

-Third paragraph: She had no signs of left ventricular dysfunction.

-Fifth paragraph: ARVD has variable manifestations and disease progression is in several phases.

-Tenth paragraph: severe instead of dangerous

Thank you very much for your collaborations,

Best regards,

Nilgün Güdücü